
SOHO Daily Meeting Minutes for Friday 18 June 2004

ANNOUNCEMENTS
-------------
* The current RHESSI default target is AR10635. When selecting this 
  target to maximize coordination w/others, use campaign number 6850.

* Visitors (high school students) in the EOF on today at 1:15 p.m.

* The B-W Parkway bridge will close to all traffic beginning this 
  evening at 7:00 p.m. It is scheduled to reopen Monday, June 21 at 
  its regular time of 5:45 a.m. The bridge will be painted during 
  this closure.

* Tomorrow is SOHO's last day with HR on 34m stations. The transponder
  swap will be on the afternoon pass, blocking out at least 1.5 hour
  at the end of the pass.

FOT REPORT
----------
Spacecraft Status: Nominal

Spacecraft Anomalies: None

Accomplished Activities: Science submode change to 6.
Executed OCD 1915.

Planned Activities: RSL Table
  Uplink delayed load: SWA040616071, VIRA40619, KEYHOLE_04171
  OCD 1919  (UVCS Safing)

Upcoming Operations: 
  16 Jun - 3 Jul     (HGA Keyhole operations)
  22 Jun             (Stationkeeping/180 deg Roll Maneuver)

Ground Anomalies: 
  169/1401 Wayward command - cause unknown - delayed commanding
             for 5 minutes, 1 command was not received by DSS24.
  169/1559 Wayward command - cause unknown - delayed commanding 
             for 8 minutes, 1 command was not received by DSS24, 
             9 commands were successfully retransmitted.

Other: N/A

SOLAR STATUS
------------
06/17 DOY 169 - 06/18 DOY 170

 EIT daily On-Line Solar Status and observations:
    http://umbra.nascom.nasa.gov/eit/plan/log_form20040618.html

 LASCO daily On-Line Solar Status and observations:
   http://lasco-www.nrl.navy.mil/observations/Todays_obs.html

 Web Page for Planning: http://www.bbso.njit.edu/arm
 
 Magnetic Maps and Heliospheric Forecasts: http://www.lmsal.com/forecast/

INSTRUMENT STATUS 

http://umbra.nascom.nasa.gov/eit/plan/log_form20040618.html
http://www.lmsal.com/forecast/
http://www.bbso.njit.edu/arm
http://lasco-www.nrl.navy.mil/observations/Todays_obs.html


-----------------
CDS:    Nominal. JOP124 and JOP176 support. Saturday: JOP124.
        Sunday: JOP124 and GIS meridian study.

SUMER:  Nominal. Not observing.

UVCS:   Nominal. At 15 UT today UVCS will enter a safe configuration 
        for the SOHO keyhole.

LASCO:  Nominal. C2/C3 synoptics at improved cadence and JOP159.
        Weekend: C2/C3 synoptics at improved cadence.

EIT:    Nominal. EIT is in bakeout through the keyhole.

MDI:    Nominal. Full disk magnetograms and dopplergrams, 
        JOP176 support today and tomorrow. 96 minute synoptic 
        magnetograms only begin on Sunday.

TRACE:  Nominal. AR10634 observations and full disk mosaics.


